Author’s response to reviews

Title: Involvement of male in antenatal care, birth preparedness, exclusive Breast feeding and immunization for child in Kathmandu, Nepal

Authors:

Dharma N Bhatta Mr. (dnbhatta@yahoo.com)

Version: 2 Date: 2 July 2012

Author’s response to reviews: see over
To
The Chief Editor
BMC Public Health

Subject: Revision of submitted manuscript.

Dear Editor-in-Chief,

Please find enclosed my revised manuscript, “Involvement of male in antenatal care, birth preparedness, exclusive Breast feeding and immunization for child in Kathmandu, Nepal” by Dharma Nand Bhatta, which was submitted for publication as a peer review in BMC public health. Major revisions are made in manuscript as per the provided comments, English language was corrected by public health professional from USA and other interpretations and discussion also corrected.

I look forward to hearing from you at your earliest convenience.

Sincerely yours’
Dharma Nand Bhatta
dnbhatta@yahoo.com